Author's response to reviews

Title: Do the socioeconomic impacts of antiretroviral therapy vary by gender? A longitudinal study of Kenyan agricultural worker employment outcomes

Authors:

- Bruce A Larson (blarson@bu.edu)
- Mathew P Fox (mfox@bu.edu)
- Sydney Rosen (sbrosen@bu.edu)
- Margaret Bii (mbii@wrp-kch.org)
- Carolyne Sigei (csigei@wrp-kch.org)
- Douglas Shaffer (dshaffer@wrp-kch.org)
- Fredrick Sawe (fsawe@wrp-kch.org)
- Kelly McCoy (mccoyk@bu.edu)
- Monique Wasunna (Africa@dndi.org)
- Jonathan L Simon (jonsimon@bu.edu)

Version: 7 Date: 13 July 2009

Author's response to reviews:

I used track changes to identify the minor edits done to the paper. I think I just deleted one "period" after my first name and changed some column sizes in a couple tables.

Thanks.